Author’s response to reviews

Title: The Effects of Isolated Ankle Strengthening and Functional Balance Training on Strength, Running Mechanics, Postural Control and Injury Prevention in Novice Runners: Design of a Randomized Controlled Trial

Authors:

Jennifer Baltich (jbaltich@ucalgary.ca)
Carolyn A Emery (caemery@ucalgary.ca)
Darren Stefanyshyn (stefanys@ucalgary.ca)
Benno M Nigg (nigg@ucalgary.ca)

Version: 2 Date: 23 July 2014

Author’s response to reviews:

Hello Dr. De Guzman

I have forwarded the ethics approval (initial 2013, renewal 2014) for the study protocol manuscript “The Effects of Isolated Ankle Strengthening and Functional Balance Training on Strength, Running Mechanics, Postural Control and Injury Prevention in Novice Runners: Design of a Randomized Controlled Trial” to BMCSeriesEditorial@biomedcentral.com.

There are no external operating grants for this research. However, I have received a student scholarship based on this research. I have attached the scholarship acceptance for the Vanier Canada CIHR Doctoral scholarship in case this is needed.

The study is currently ongoing and actively recruiting, collecting and analyzing the data.

There are no related articles that have been published related to this research to date.

Please let me know if there is any additional documentation that I need to provide.

Thank you

Jennifer